Author’s response to reviews

Title: Nasopharyngeal cancer mimicking Otitic Barotrauma in a resource challenged centre: a case report and literature review.

Authors:
Adekunle Daniel (kunle_d2002@yahoo.com)
Ayotunde J Fasunla (ayofasunla@yahoo.com)

Version: 4 Date: 21 August 2011

Author’s response to reviews:

Dear Editor In-Chief,

JMCR

Dear Sir/Ma,


Thank you for the suggestions and the corrections.

We have asked a colleague whose native language is English to review the manuscript. The changes are highlighted in red.

1. Line 11 under the introduction: ‘common benign ear pathologies’ has been changed to common benign ear diseases.

2. Line 11 under the discussion: ‘within the nasopharynx and paranasopharyngeal extension’ has been changed to ‘within the nasopharynx and with paranasopharyngeal extension.’

Thank you once again

Yours faithfully,

Dr Adekunle Daniel.